---
sidebar_position: 1
id: "index"
title: "Home"
---

import {Card, CardGroup} from '@site/src/components/card.tsx';

## Welcome to Wave Terminal

Wave is an [open-source](https://github.com/wavetermdev/waveterm) terminal that adds the ability to launch graphical widgets, controlled and integrated directly with the CLI.  We support MacOS, Linux, and Windows ([Downloads](https://waveterm.dev/download)).

**These docs are incomplete**, but we're working on them.  If you have a question, please feel free to ask us in [Discord](https://discord.gg/XfvZ334gwU).  If you'd like to file a bug/enchancement, please
use [Github Issues](https://github.com/wavetermdev/waveterm/issues).  These docs are also open-source and we do accept PRs in our [docs repo](https://github.com/wavetermdev/waveterm-docs)

References:

* [Widgets](./widgets)
* [Configuration](./config)
* [Key Bindings](./keybindings)
* [wsh command](./wsh)
* [Telemetry](./telemetry)
* [FAQ](./faq)

<!--
## Features

<CardGroup>
	<Card href="features/preview" icon="fa-file-magnifying-glass" title="File Previews" description="Preview widgets let you view markdown, images, video, CSVs, and more inline." />
	<Card href="features/remotes" icon="fa-server" title="Reliable Remote Connections" description="Automatic reconnection, remote file previews, and more." />
	<Card href="features/ai" icon="fa-sparkles" title="Integrated AI" description="Inline AI Chat Windows hooked up to any LLM (local or remote)." />
	<Card href="features/browser" icon="fa-globe" title="Inline Web Browser" description="Access online content without switching context." />
	<Card href="features/charts" icon="fa-chart-simple" title="Data Visualization" description="Create graphs of system stats or custom data piped directly from CLI programs." />
</CardGroup>
-->

## Links

* Homepage -- https://waveterm.dev
* Download -- https://waveterm.dev/download
* Discord -- https://discord.gg/XfvZ334gwU
* Main Github (source code) -- https://github.com/wavetermdev/waveterm/
* Docs Github (powers this site) -- https://github.com/wavetermdev/waveterm-docs/

## Looking for WaveLegacy documentation?

WaveLegacy docs can be found at [legacydocs.waveterm.dev](https://legacydocs.waveterm.dev).
